Author's response to reviews

Title: Water sources as reservoirs of Vibrio cholerae O1 and non-O1 strains in Bepanda, Douala (Cameroon): relationship of isolation and physico-chemical factors.

Authors:

Jane-Francis T.K. Akoachere (jakoachere@yahoo.com)
Christelle K.P. Mbuntcha (cpulcherie2000@yahoo.com)

Version: 5 Date: 27 March 2014

Author's response to reviews: see over
27th March, 2014

The Editor
BMC Infectious Diseases
Dear Sir,

**Submission of Revised Manuscript (ID: 1215834269107529)**

We have addressed all the concerns raised by the reviewers after their second review and are submitting our revised manuscript entitled ‘Water sources as reservoirs of *Vibrio cholerae* O1 and non-O1 strains in Bepanda, Douala (Cameroon): relationship of isolation and physico-chemical factors’. Responses to reviewers are submitted as separate files. The following corrections have been done on the manuscript:

1. The manuscript has been edited and spelling and spacing errors as well as citations corrected.
2. Some sentences have been revised and quantitative results which were presented in the results section and then repeated in the discussion deleted from the discussion.
3. We have provided clarifications with regards to methodology.

We sincerely appreciate the comments and contributions of the reviewers

Yours sincerely,

Jane-Francis Akoachere, Ph. D